

Contents lists available at ScienceDirect

Women and Birth

journal homepage: www.sciencedirect.com/journal/women-and-birth





Women's experiences of intermittent auscultation fetal monitoring in labour: A qualitative study

Jennifer MacLellan ^{a,*}, Mo Ade ^b, Bev Fitzsimons ^c, Sara Kenyon ^d, Sumayya Mulla ^b, Catherine Pope ^e, Julia Sanders ^f, Rachel Rowe ^a

- a National Perinatal Epidemiology Unit, Nuffield Department of Population Health, University of Oxford, Old Road campus, 01865 289700, Oxford OX3 7LF, UK
- b Expert by experience contributor, UK
- ^c Point of Care Foundation, 99 Gray's Inn Road, London WC1X 8TY, UK
- d Institute of Applied Health Research, College of Medical and Dental Sciences, University of Birmingham, Edgbaston, Birmingham B15 2TT, UK
- e Nuffield Department of Primary Healthcare Sciences, University of Oxford, Radcliffe Quarter, Woodstock Road, Oxford OX2 6GG, UK
- ^f School of Healthcare Sciences, Heath Park campus, Cardiff University, Cardiff CF14 4XN, UK

ARTICLE INFO

Keywords: Midwifery Fetal monitoring Women's experiences

ABSTRACT

Background: Internationally, intermittent auscultation (IA) is recommended for monitoring the fetal heart rate during labour and birth for women with uncomplicated pregnancies. IA can identify changes in the fetal heart rate that may indicate the need for additional care or intervention. IA is a central facet of midwifery practice, but there is little evidence about women's experience of IA.

Aim: Our study aimed to explore women's experiences of IA in the UK.

Methods: Between February and May 2023, 23 women were recruited through social media, service user organisations and charities, for a single episode, online interview or focus group with informed consent. Audio recordings were transcribed and thematically analysed.

Findings: The analysis constructed two over-arching themes: 'Choice takes work' and 'Impact of IA monitoring on the labouring woman'.

Discussion: Women reported a lack of informed decision-making in relation to intrapartum fetal monitoring. The experience of our participants showed that not all women for whom IA would be recommended according to current clinical guidelines were offered it, while others experienced IA without understanding its function or the availability of other options.

Conclusion: Limited antenatal communication from maternity care professionals about fetal monitoring in labour has an impact on women's opportunities to make informed decisions. A committed approach to informed decision making in the antenatal period could reduce practice variation and better support midwives to support women in their birth choices.

Significance Statement

Problem or issue

Intermittent auscultation (IA) is a central facet of midwifery practice, but there is very little evidence about women's experience of IA.

What is already known

IA is recommended for fetal monitoring during labour for women

with uncomplicated pregnancies. IA allows increased mobility in labour compared with continuous monitoring, and potentially better communication and holistic assessment of the labouring woman.

What this paper adds

Limited communication about fetal monitoring practice in labour from maternity care professionals in the antenatal period has an impact on women's opportunities to make informed decisions about monitoring.

https://doi.org/10.1016/j.wombi.2024.101805

Received 30 April 2024; Received in revised form 19 August 2024; Accepted 20 August 2024 Available online 29 August 2024

^{*} Corresponding author.

E-mail address: jennifer.maclellan@npeu.ox.ac.uk (J. MacLellan).

@RachelRowe3 (R. Rowe)

1. Background

The aim of fetal monitoring during labour is to identify changes in the fetal heart rate and pattern that may indicate inadequate oxygen supply to the baby so that additional care or intervention, such as instrumental vaginal birth or caesarean birth, can be implemented to improve outcomes for the baby [14]. The World Health Organization, International Confederation of Midwives (ICM), and the International Federation of Gynaecology and Obstetrics recommend 'intermittent auscultation' (IA) for fetal monitoring in uncomplicated pregnancies during labour and birth, and in settings where no alternative approach to fetal monitoring is available [5,11,12,35]. If there are no identified risk factors for fetal compromise in labour, IA is recommended in preference to continuous electronic fetal monitoring (CEFM) [1,3,17]. Moreover the use of CEFM, especially for low risk labours is contested, indeed at least one leading obstetric researcher has argued that "more we use it [CEFM], the more harm we do" [16]. The research evidence shows that CEFM in 'low risk' women is associated with an increase in intervention during labour and birth without improving outcomes [22, 30,91.

IA involves locating and listening to the fetal heart rate immediately after a contraction using a pinard or hand held Doppler ultrasound device. There is limited evidence recommending one device over another [14]. During the first stage of labour, it is recommended that IA is carried out every 15-30 minutes, with United Kingdom (UK) guidelines and those of the ICM recommending every 15 minutes [24]. In the second stage of labour, most guidelines recommend listening at least every five minutes [5,17]. The midwife will auscultate the fetal heart beat for at least a full minute [17], recording the presence of increases (accelerations) or decreases (decelerations) in the heart rate. If, on intermittent auscultation, there is an increase in the fetal heart rate baseline (as plotted on the partogram) of 20 beats a minute or more from the start of labour, or a deceleration is heard, full review taking into account the whole clinical picture and potential further action is recommended [12, 17]. Further action might include a period of more frequent listening-in, transfer to CEFM and/or intervention to expedite birth [17].

IA is a central facet of midwifery practice around the world. In the UK, national enquiries have raised concerns about midwifery IA practice [15,23], and a programme of research (Listen2Baby) is underway aiming to provide research evidence to support the co-design of a 'toolkit' to improve midwifery IA practice (researchregistry ref:8463). Women's experience of IA is an important consideration as an indicator of quality of care [20], but there is very little research evidence. A systematic review of ten studies about women's experience of fetal monitoring, conducted between 1976 and 2013, documented that, compared with CEFM, IA allowed increased mobility in labour, closer proximity and engagement with women by the midwife, highlighted by some women as important, and potentially better communication and holistic assessment of the woman and baby's clinical condition [29]. However, only two of these studies included women's experiences of IA and only one of these was conducted after 1985. The most frequent procedures during pregnancy and birth, including fetal monitoring, are those that are least likely to have been discussed and/or informed consent sought [31,4]. As the first step in the Listen2Baby project we aimed to capture the neglected perspective of women's experience of IA.

2. Method

Between February and May 2023, we aimed to recruit women who had experienced IA during labour through advertisements posted on social media, and through service user organisations and charities involved in the project or known to members of the research team. All respondents were sent further information about the study with an interview consent form and invited to take part in a focus group or

individual interview, whichever they preferred.

A topic guide was developed by RR, JM and BF in response to the study research question. The guide was piloted with two women with lived experience of IA and used to create a broad structure for the interview, with flexibility for the participant to develop and expand on the topics of particular importance to their experience. Data were collected through single episode interviews or focus groups with informed consent. These were conducted online or by telephone by an experienced qualitative researcher (JM). Audio recordings were transcribed verbatim, checked against the original recording, and loaded into the data analysis software, NVivo 13 [13].

We analysed the data thematically as data collection progressed [6]. The primary researcher (JM) coded the transcripts and developed core categories. These were interpreted into themes in discussion with the wider multidisciplinary research team (RR, JS, SK, CP, SM, MA). The team included two lay members with lived experience of IA in their labours. In line with their preferences about terminology, and those of the women who took part in interviews, we refer to 'fetal' heart rate monitoring in this paper, but also use the word 'baby' rather than 'fetus' when reporting what the woman said.

We purposively sampled for diverse experiences and collected participant characteristics to ensure variation in our sample. During the interviews, it became clear that not all participants had experience of IA during established labour, although they had earlier confirmed that they had IA in labour. Those who had CEFM had not been aware that there were different fetal monitoring options for labour. We coded the data from these participants separately, but following careful consideration within the research team, we then incorporated it into the main dataset as we found it added explanatory value to our categories about the (lack of) information sharing and choice-making discussions. We were only able to recruit one partner and so excluded their data from this analysis. Sampling continued until data saturation was agreed by the wider team to have been reached [28].

3. Results

We spoke to a diverse sample of 23 women who had given birth in the preceding 3 years. We conducted 16 individual interviews, one two-person interview, and a focus group of six participants supported by an interpreter. Interview discussions lasted between 20 and 65 minutes (average 36 minutes) with the focus group lasting 74 minutes. Selected participant characteristics are summarised in Table 1.

Table 1: Participant characteristics

Twelve of the women in our sample described their most recent pregnancy as being classified by their maternity provider as at 'low risk' of complications and 11 were classified as 'higher risk'. Six women gave birth at home, six in a midwifery unit and 11 in an obstetric unit. In terms of their fetal monitoring experience in their most recent pregnancy, one woman had no fetal monitoring, 12 had IA only, four had IA

Table 1Participant characteristics.

Characteristic		Number of participants
Age	26-30	8
	31-35	12
	36-40	3
Ethnicity	White British	13
	Mixed Black and Other	2
	South Asian	6
	British Asian	2
Education	Graduate	16
	High school	5
	Missing	2
Number of births experienced	One	12
	Two	4
	Three	3
	Four or more	4

and CEFM, and six had only CEFM.

The analysis constructed two over-arching themes, each explained by 2 sub themes:

- 1. Choice takes work:
 - a. Knowing about IA informs choice
 - Not knowing or not remembering being told about fetal monitoring constrains choice
- 2. Impact of IA on the labouring woman
 - c. IA as disruptive
 - d. IA as reassuring

3.1. Choice takes work

Some of the women we spoke to described being unaware of different fetal monitoring methods, or the possibility of discussing different options and choices with their care provider. They described accepting the recommendation or practice of the maternity care team on admission in labour, often without any discussion. However, some of the women we spoke to described searching for information about birth place choices and that this was where they learned of the different types of monitoring most common in different settings. They described this information and the support from their midwifery care team, as directing their choices.

3.1.1. Knowing about IA informs choice

Those participants who knew about fetal monitoring options in labour described how they did their own information gathering about how they would like to be supported in their birth. This could be as a result of a previous poor experience, and looking for alternative options, or to inform choices in a first time birth experience about how they would like to be supported. They then brought this information to their antenatal appointments to discuss with the midwife.

".because I had a bad experience, I had to get my head around it before the third to come to terms with it. So I'd actually done quite a bit of reading around how to get what I wanted for a birth, and one of the things that came up a lot was having intermittent monitoring. I wanted a very hands-off, leave me alone [birth]" Alison, experience of IA in labour, four or more births.

"I started looking at birth study research and went, 'oh actually, maybe a home birth is for me.' I looked at the 'cascade of intervention' and knew that, 'OK, so this is what I want to try and avoid, I want to try and have as little intervention as possible, and little disruption to the natural setting for my birth, and I wanted my birth day to be right." Sally, experience of IA in labour, one birth.

If participants had opted for a home birth, they described having preparatory meetings, visits or webinars where the activities of the midwife in labour were presented and discussed by their care provider. This included discussion of intermittent fetal monitoring, what it is for, how it is done, and what happens if there are any concerns. This information giving was described by women as contributing to them feeling informed about their birth.

"the home birth midwife team did an online webinar that said what to expect at your home birth, so that mentioned the monitoring, so I knew it was gonna happen and I knew they were gonna be doing that, but I guess that's a positive thing that I knew to expect them to monitor it and how often." Anna, experience of IA in labour, one birth.

Support for women's choice around fetal monitoring and associated care was variable across our sample. Participants described experiences of when this communication worked well and they felt supported.

"having had conversations with my midwife, she said, "There will be times where I will sit you down and I will say to you in this situation: 'you are better being in hospital', [...] and I think I really respected that honesty, actually I don't know all of the things that can go wrong and the things that they're on the lookout for, so in that situation, I definitely would have trusted, her advice" Alice, experience of IA in labour, one birth.

"the matron was happy to sign me off for a home birth, or a birth centre birth [that would be monitored using IA] so it gave me a bit of confidence" Grace, experience of IA in labour, three births.

However, some practitioners were described as more risk-averse than others. This could result in inconsistency in both messaging and potential care decisions, which caused confusion and anxiety for the woman

"although I'd had it all signed off [IA and Midwife Led Unit], I did end up going for an induction [...] when it got to the point where I needed to be transferred to the midwife led unit, the delivery suite did get quite antsy with my poor midwife, and tried to tell that because I had a section before, she needed to send me down to delivery [for CEFM monitoring], she said 'well no, it's written in the notes that have been signed off by so-and-so, and I'm taking her upstairs, so I'm sorry" Alison, experience of IA in labour, four or more births.

Some participants described being classified as 'higher risk' due to previous caesarean birth, previous pregnancy loss, IVF conception or higher weight. They described their choices to give birth outside of an obstetric unit or to have IA instead of the recommended continuous CEFM as 'outside of guidance' [guidance for uncomplicated pregnancies that are recommended for midwifery led care and intermittent fetal monitoring].

These participants felt they had to negotiate to have IA 'outside of guidance'. They described how they were accepting of more intensive CEFM monitoring if the IA raised concerns about the fetal experience of labour but they did not want to start their labour with the restrictions associated with CEFM. Their request for an active, mobile labour, which IA was felt to support, was based on their belief that IA worked (it was effective in revealing red flags that might warrant a change in management), midwives were competent to interpret and act on the IA findings, and that the birthing woman's choices considered the mother and her unborn baby as an interdependent unit.

"My big argument was, I don't want to start at intervention level three when actually we will be perfectly fine on the base level." Mary, experience of IA in labour, three births.

"that goes to show how effective the intermittent was, although it wasn't a continuous thing, it still picked it up enough that there was a problem without being strapped to a monitor, which is what I didn't want, I wanted to be able to move, I wanted to have the freedom and, there is definitely a time and a place for continuous monitoring, but the intermittent was definitely effective enough to pick up a problem before it was a problem really, 'cause he was absolutely fine and it still picked it up well in advance" Alison, experience of IA in labour, four or more births.

Among women who knew about IA these conversations appeared to reflect local practice and midwives' adherence to guidelines around recommended eligibility for planning place of birth rather than individualised assessment and care. This led to some challenging conversations between participants and practitioners that participants found stressful.

"I would have liked a list of the evidence for and against what I was being told, or advised against, rather than just being told: 'this is how we do it,' I'd have never have thought with my first that I would have gone against all guidance with my third [IA, Waterbirth, VBAC] and it was the evidence that I'd looked at" Alison, experience of IA in labour, four or more births.

"we had IVF, and that was kind of thrown at me as, "Oh well, we don't recommend it [homebirth and IA] in IVF," but no real kind of facts or research to back that up" Sarah, experience of IA in labour, one birth.

While some participants shared strong preferences for their birth place and the fetal monitoring associated with that care pathway, all participants requested communication about what was being done in labour and why, and how it might inform care decisions. Participants reported distress when recommendations were framed as usual practice, removing the individuality of their experience, or when they experienced inconsistent support. Participants accepted that the maternity care team were experts in birth, but placed themselves as experts in their body.

3.1.2. Not knowing or not remembering being told about fetal monitoring constrains choice

Some of the women we spoke to could not recall fetal monitoring options or practice being discussed before or during their labour.

"once I went into the delivery suite they attached the monitors round me and that's how it stayed for the rest of my labour. Yeah, I don't remember a conversation about whether that's what I wanted, or whether that was necessary." Lucy, experience of IA in labour, one birth.

"I had written in my birth plan actually just to talk me through any intervention that you are doing, just for consent, but also just for peace of mind and reassurance [...] it didn't really particularly stick in my mind that they had explained the intermittent monitoring and why they were doing it." Isa, experience of IA in labour, one birth.

"I think it's one of those things that as a professional you know what the Doppler is and you know why you are doing it, and that's just part of the procedures, and people assume that the women understand exactly what this is" Margaret, experience of IA in labour, three births.

Lack of communication about fetal monitoring was described by this group of participants as increasing their anxiety during the birth as they didn't understand what was going on. This contributed to confusion and reduced the woman's sense of control.

"Yeah, every time when I had that monitor with me, and like half an hour they did, and 10-minute rest, again did, but I don't know why it was happening." Padma, no experience of IA in labour, four or more births

"so I definitely did panic when the student midwife couldn't find the heartbeat, but if someone had said to me beforehand, 'oh, sometimes we won't be able to find it and you'll need to change positions,' and that kind of thing, and it would have just been a normal thing, rather than I was like, "Oh my God, why isn't there a heartbeat, what's happening?" So yeah, I think a bit more prep about what that's about and what it's for beforehand would been helpful." Sarah, experience of IA in labour, one birth.

"I feel like I was more in control because they were explaining things to me; whereas in the second one, I honestly, I remember there were times I was looking at my husband like, "What is going on?" [.] I started getting really worried. I know my husband was getting worried thinking, 'what is happening, why are they all rushing around, and why's this not being communicated to us?' Maya, experience of IA in labour, two births.

Participants talked about the need for midwives to check that women are aware of and understand their options, and not to make assumptions about women's knowledge. They described how information was needed to make a choice, and emphasised that every woman's experience is different.

"when we go, we shy and don't know what to ask the midwife, if the patient ask the thing – it's good, but if patient don't ask- they should discuss about that topic" Padma, no experience of IA in labour, four or more births.

"I think that midwife care needs to be tailored individually because everyone wants something different [...] I feel like I made irrational choices, but just based on anxiety, or because I didn't know what to expect." Aarya, no experience of IA in labour, one birth.

"it's checking people's understanding is probably the most important thing, because it's one thing to give someone the information, it's another thing to check how much they've taken in, and for me that's what felt light touch, is they've given me the information but no one's checked how much I've taken in." Isa, experience of IA in labour, one birth.

Participants described how conversations with maternity care professionals were more frequently framed as informing women what will happen, rather than as shared decision-making and informed consent.

"I went in with the mindset of what the midwife says, is what we did" Alison, experience of IA in labour, four or more births.

"rather than an explanation of what the options were, it was more informing me what would happen and I would have had to question that if I'd not wanted that" Sarah, experience of IA in labour, one birth.

While many participants felt comfortable and confident to follow the midwife's recommendations, there was a universal request from women that their opinion be sought and their decisions be respected.

3.2. Impact of IA on the labouring woman

The impact of IA on the labouring woman was informed by two contrasting sub themes: 'IA as disruptive' and 'IA as reassuring'. For the participants who had made clear choices for their birth, IA monitoring symbolised a 'midwifery led care' package. When participants were asked about their experience of IA in labour, responses focused on the impact of IA practice on the labouring woman herself. The accuracy and reliability of the method and the midwives' ability to interpret the data was not questioned. The participants in our study had variable responses to IA, with some finding that it disrupted their labour while others found it reassuring.

3.2.1. IA as disruptive

Some participants described how IA disrupted their labour as they were expected to communicate with the midwife about their contractions or to change position to enable the midwife to carry out the IA.

"for quite a lot of labour I shut myself in the bathroom, 'cause I felt like that was a really dark cosy space where I felt safe, and I remember feeling irritated when they kept knocking on the door to come in and monitor baby, and then on occasion obviously they would put the Doppler on when a contraction was coming and I'd had have to say, "Oh, I'm having a contraction, don't do that," but obviously the interruption felt really annoying in terms of having to verbalise that" Sarah, experience of IA in labour, one birth.

"It was disempowering to feel reliant on a machine to tell me how my birth was going and disconnected me from my body and baby and birth. Another negative on the intermittent auscultation is that it made me watch the clock as I knew 15 minutes had passed each time" Bella, experience of IA in labour, two births.

In a few cases, midwives were described as being more concerned with the fetal monitoring practice than being mindful to the comfort of the woman. This created moments of tension in the care relationship and may have contributed to the perception of IA as disruptive.

"my movements were basically disrupting the monitoring, it was more about the monitoring than my comfort, is what it felt like" Fiona, experience of IA in labour, four or more births.

"The other thing I'd say is actually it was quite uncomfortable having them listening in: some of them really dug that thing in, and I think you might... 'oh, it might be a little bit uncomfortable,' but actually that was quite uncomfortable, and so making sure that people are OK with that digging it in." Margaret, experience of IA in labour, three births.

A lack of communication about how and why midwives do fetal monitoring, and how the birthing woman is experiencing the IA can impact birthing women's experience. Some participants shared how they would work with the midwife to minimise the disruption of IA.

"I think, the baby must have been coming down the birth canal, I was basically giving birth and it felt excruciating, I do you know, I mean all she did was probably gently touch me, but it was really not nice, so I just said, "Oh, please stop," and she did" Grace, experience of IA in labour, three births

3.2.2. IA as reassuring

For many participants, IA monitoring was described as reassuring, with the midwife's consideration for the woman's comfort and her skills being valued. Participants described working with the midwife in the flow of their labour to signify the end of a contraction or altering position to enable the midwife to access her abdomen to hear the fetal heart.

"I suppose I found it reassuring that everything was fine. But for me it was a happy medium because I felt like I had that reassurance, but it wasn't intrusive, and they were quite respectful and if I'd have said, 'not right now,' they would have stepped back" Grace, experience of IA in labour, three births.

"they explained to me that what they would do is when I felt a contraction coming on, if I could let them know, they would monitor and listen in to baby, and that they would just keep doing that, and they could do that once I was in the water as well. So yeah, I was happy with that, I was happy to have the monitoring 'cause I wanted to make sure that baby was doing OK, but also really happy that I didn't have to be stuck to the bed, that I could move around, and that they could still do that while I was in the water as well" Anna, experience of IA in labour, one birth.

Most participants described midwives as communicating findings or changes in the fetal monitoring schedule in calm, reassuring ways. This was described by participants as enabling them to feel informed and in control of their experience.

"They just kept me informed, even though everything seemed to be going okay, just to let me know, 'yeah it sounds good', that was positive." Anna, experience of IA in labour, one birth.

These findings emphasise the individuality of women's experience of labour and the skills and flexibility of the midwife required to work with the woman, her preferences and embodied experience of labour.

4. Discussion

In this study we explored women's experience of IA as part of a wider project aiming to improve midwifery practice of IA in the UK. We found that information about fetal monitoring was not routinely shared with our participants antenatally or discussed during labour. While women often felt supported when they shared their choices with practitioners, this could involve difficult conversations if these choices were considered as 'outside of guidance'. The experiences of our participants suggested that not all women for whom IA would be recommended (as per guidance) were offered this during pregnancy or in labour.

Informed choice is based on the availability and receipt of relevant and balanced information to allow the individual to weigh the risks and benefits according to their own values and interpretations) [36]. Women and practitioners have reported a lack of informed decision-making in maternity care, with options not always being presented to women [33, 36,4,7]. This includes discussion of fetal monitoring [31,4]. If women are not given information about fetal monitoring and birth place choices, they risk losing control over the medical decisions facing them [2]. Furthermore, women and practitioners have been described as acculturated to accept CEFM as the expected method of fetal monitoring in labour, independent of evidence-based eligibility criteria for intermittent monitoring [1,10,17,30]. Local interpretation of guidelines allows for local practice culture to influence the offer of fetal monitoring options or choices. While this can allow some flexibility for midwives and women, there is also the potential to cause confusion if women are not fully informed of their options and supported to advocate for their choice. For example, in relation to fetal monitoring during vaginal birth after caesarean (VBAC) or during the labour of women with higher weight, the RCOG green top guideline recommends CEFM [26], but some local guidelines indicate that either CEFM or IA may be possible

We found parallels with women's experience of choice making for IA monitoring, with research about women's experiences of planning place of birth. Evidence suggests that women who prefer to plan birth outside of hospital, at home or in a midwifery unit, often have to actively seek out information themselves to inform their birth place choice [8]. Midwives may restrict choice in relation to birth place by not presenting the alternatives or not being encouraging when women express a preference [18]. Conversely, women felt supported when they were given information by their midwife and supported or offered different birth place options as a choice [8]. Some women in our study saw fetal monitoring practice in general, and IA in particular, as part of a package of care, inextricably linked with planned place of birth and a care pathway which was supportive of an intervention-free labour and birth. These women actively sought out or requested IA; for some this meant they were giving birth 'outside of guidance'. Maternity professionals have an obligation to provide accurate and detailed information to support women's decision making. If women who do not have a guideline identified risk factor have a right to decide on the form(s) of fetal heart rate monitoring to be used in their labour, women with risk factors have the same rights and professionals have the same obligations towards them. However, the experience of our participants showed that not all women were informed about fetal monitoring options, that women who met criteria for IA were not always offered it, while some had it without really knowing what it was or what it was for.

Ultimately, women will always have individual responses to fetal heart monitoring. For many it is reassuring; others find it disruptive. Some would rather it were done more often, or may prefer continuous fetal monitoring, and others less. Some will decline fetal monitoring altogether. This means that midwives need to inform women about the rationale for recommending IA or CEFM, find out about women's individual choice beforehand, and check in with women about their choices during labour. In carrying out IA, midwives balance the need to follow practice guidelines, while minimising disruption to women's labour [21]. One of the key recommendations of the Royal College of Midwives Professional Briefing on caring for women seeking choices that fall outside guidance is that 'Midwives should support the woman to make informed decisions and suspend judgements on those decisions; clinicians are not required to 'approve' women's informed choices' [25], and yet some of our participants described the need for 'sign off' for care that differed from local guidance. It is recommended that care providers work with women in the antenatal period to include discussions about fetal monitoring in labour otherwise not all women will be in a position to make an informed decision for their labour. Our research has shown that while IA can be disruptive to a woman's labour, this can be mitigated through discussion of why IA is being done, how it is done and

what the findings could mean for the woman and her baby. This informed approach allows the woman and midwife to work together with the woman's labour to maximise safety and comfort. Ideally these conversations would be recorded in the birth plan or the woman's Personalised Care and Support Plan (PCSP) [19,34].

5. Strengths and Limitations

Our study uniquely explored women's experiences of IA in labour within UK maternity services, but may also help inform IA practice in other countries. Our sample was diverse in terms of participant characteristics and experiences. Our principal limitation was the use of social media and service-user organisations to recruit our participants, with the consequence that most participants were well-informed about fetal monitoring options and had actively made a choice for IA or a birth place option that would support this type of fetal monitoring. We purposively sampled to ensure that we heard the experiences of women without a strong fetal monitoring preference during labour, but our data may overrepresent those who explicitly did not want CEFM. A small number of our participants had no direct experience of IA, which was not our intention when we planned this study. Invitations to take part in the study, and information provided about the aim of the study, clearly described the kind of fetal monitoring experience that we were interested in. Nevertheless, it became apparent that six of our participants did not have experience of IA, although two of these women should have been recommended for IA according to current guidelines. The experience of these women, in terms of lack of awareness and discussion of the different types of fetal monitoring with their midwife, reinforces the themes identified in our analysis and was informative in the broader context about awareness and choice, so we took the decision to include their data in the study.

6. Conclusion

Our participants described limited communication about fetal monitoring practices in labour from maternity care professionals in the antenatal period. This had an impact on the information and support they received and their opportunity to make informed decisions. Participants with a strong birth place or fetal monitoring preference described challenging a lack of evidence for local restrictions on their choice. Some participants were better supported in making choices than others and this may reflect the strength of local practice cultures or the individual providers of care. An approach to informed decision making in the antenatal period that supports midwives to have iterative conversations with women about their values and preferences could reduce practice variation and better support midwives to support women in their birth choices.

Funding statement

This study is funded by the National Institute for Health and Care Research (NIHR) Health and Social Care Delivery Research (HSDR) Programme (NIHR134306). The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care.

Author agreement

We, the authors state that this article is our original work. It has not received prior publication and is not under consideration for publication elsewhere. All authors have seen and approved the manuscript being submitted. The authors abide by the copyright terms and conditions of Elsevier and the Australian College of Midwives. Furthermore, the authors have no conflicts of interest to declare

Ethical Statement

This study has received ethical approval from University of Oxford Central University Research Ethics Committee [ref: R76969_RE001] on 01/12/22.

CRediT authorship contributions statement

RR, JS, SK, BF, CP conceptualized the study; RR secured funding and designed the Methodology with BF, CP. The project was administered and data collected by JM. Formal analysis was undertaken by JM, CP, RR, with contributions of SK, JS, BF, MA, SM. The project was supervised by RR. JM wrote the original draft with all authors contributing to review and editing. All authors approve the final manuscript.

Conflict of Interest statement

The authors have no conflicts of interest to declare

Acknowledgements

We would like to thank the women for their time and sharing of their experiences for this study and the charities who supported us:

References

- [1] Z. Alfirevic, D. Devane, G.M.L. Gyte, A. Cuthbert, Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour, Cochrane Database Syst. Rev. 2 (2017) CD006066, https://doi.org/ 10.1002/14651858.CD006066.pub3, pmid:28157275.
- [2] A.G. Andal, Politics of birth places: examining the ethics of birth place choices, JAHR – Eur. J. Bioeth. 11 (1) (2020) 61–82.
- [3] K. Anderson, J. Salera-Vieira, E. Howard, The evidence for intermittent auscultation, J. Perinat. Neonatal Nurs. 37 (3) (2023) 173–177, https://doi.org/ 10.1097/JPN.0000000000000754.
- [4] E. Annandale, H. Baston, S. Beynon-Jones, L. Brierley-Jones, A. Brodrick, P. Chappell, et al., Shared decision-making during childbirth in maternity units: the VIP mixed-methods study, Health Soc. Care Deliv. Res. 10 (36) (2022). (https://www.journalslibrary.nihr.ac.uk/hsdr./FPFP4621#/s4).
- [5] E. Blix, R. Maude, E. Hals, S. Kisa, E. Karlsen, E.A. Nohr, A. de Jonge, H. Lindgren, S. Downe, L.M. Reinar, M. Foureur, A.S.D. Pay, A. Kaasen, Intermittent auscultation fetal monitoring during labour: A systematic scoping review to identify methods, effects, and accuracy, Jul 10, PLoS One 14 (7) (2019) e0219573, https://doi.org/10.1371/journal.pone.0219573.
- [6] V. Braun, V. Clarke, Using thematic analysis in psychology, Qual. Res Psychol. 3 (2) (2006) 77–101.
- [7] D. Coates, N. Donnolley, M. Foureur, A. Henry, Women's experiences of decision-making and attitudes in relation to induction of labour: a survey study, Women Birth 34 (2) (2021) e170–e177, https://doi.org/10.1016/j.wombi.2020.02.020.
- [8] K. Coxon, A. Chisholm, R. Malouf, R. Rowe, J. Hollowell, What influences birth place preferences, choices and decision-making amongst healthy women with straightforward pregnancies in the UK? A qualitative evidence synthesis using a 'best fit' framework approach, Mar 31, BMC Pregnancy Childbirth 17 (1) (2017) 103, https://doi.org/10.1186/s12884-017-1279-7.
- [9] D. Devane, J. Lalor, S. Daly, W. McGuire, A. Cuthbert, V. Smith, Cardiotocography versus intermittent auscultation of fetal heart on admission to labour ward for assessment of fetal wellbeing, Cochrane Database Syst. Rev. 1 (1) (2017) CD005122, https://doi.org/10.1002/14651858.CD005122.pub5.
- [10] L. Heelan, Fetal Monitoring: Creating A Culture of Safety with Informed Choice, J. Perinat. Educ. 22 (3) (2013) 156–165, https://doi.org/10.1891/1058-1243.22.3.156.
- [11] International Confederation of Midwives. Use of Intermittent Auscultation for Assessment of Foetal Wellbeing during Labour. International Confederation of Midwives; 2017. Available from: (https://www.internationalmidwives.org/assets/ files/statement-files/2018/04/eng-use_intermittend_auscultation.pdf).
- [12] D. Lewis, S. Downe, FIGO consensus guidelines on intrapartum fetal monitoring: intermittent auscultation, Int. J. Gynaecol. Obstet. 131 (2015) 9–12, https://doi. org/10.1016/j.ijgo.2015.06.019.
- [13] Lumivero (2020) NVivo (Version 13. R1) (www.lumivero.com).
- [14] R. Martis, O. Emilia, D.S. Nurdiati, J. Brown, Intermittent auscultation (IA) of fetal heart rate in labour for fetal well-being, Cochrane Database Syst. Rvws (2017), https://doi.org/10.1002/14651858.CD008680.pub2.
- [15] Maternity and Neonatal Safety Investigations Programme National Learning Report. Factors affecting the delivery of safe care in midwifery units. (https://mn si-2zor10x7-media.s3.amazonaws.com/production-assets/documents/ 20240508-MNSI-Delivering-safe-care-in-midwifery-units.pdf) (accessed online 18.07.24).

- [16] E. Mullins, C. Lees, P. Brocklehurst, Is continuous electronic fetal monitoring useful for all women in labour? BMJ 359 (2017) j5423, https://doi.org/10.1136/bmj. i5423
- [17] National Institute for Health and Care Excellence, Intrapartum care: Fetal monitoring in labour. NICE guideline CG229, NICE, London, 2022. Available from: https://www.nice.org.uk/guidance/cg229).
- [18] J. Naylor Smith, B. Taylor, K. Shaw, et al., I didn't think you were allowed that, they didn't mention that.' A qualitative study exploring women's perceptions of home birth, BMC Pregnancy Childbirth 18 (2018) 105, https://doi.org/10.1186/s12884-018-1733-1.
- [19] NHS England (2021) Personalised Care and Support Planning Guidance. Guidance for Local Maternity Systems. (https://www.england.nhs.uk/wp-content/upload s/2021/03/B0423-personalised-care-and-support-planning-guidance-for-lms.pdf) (accessed 05/10/2023).
- [20] M. Redshaw, C.R. Martin, E. Savage-McGlynn, S. Harrison, Women's experiences of maternity care in England: preliminary development of a standard measure, BMC Pregnancy Childbirth 19 (2019) 167.
- [21] R. Reed, J. Rowe, M. Barnes, Midwifery practice during birth: ritual companionship, Women Birth 29 (3) (2016) 269–278, https://doi.org/10.1016/j. www.bi.2015.1.002
- [22] I.K. Rosset, K. Lindahl, E. Blix, A. Kaasen, Recommendations for intrapartum fetal monitoring are not followed in low-risk women: a study from two Norwegian birth units, Sex. Reprod. Healthc. 26 (2020) 100552, https://doi.org/10.1016/j. srbc.2020.100552
- [23] R. Rowe, E.S. Draper, S. Kenyon, C. Bevan, J. Dickens, M. Forester, R. Scanlan, D. Tuffnell, J.J. Kurinczuk, Intrapartum-Related Perinatal Deaths in Births Planned in Midwifery-led Settings in Great Britain: Findings and Recommendations from the ESMiE Confidential Enquiry, BJOG 127 (13) (2020) 1665–1675, https://doi. org/10.1111/1471-0528.16327.
- [24] Royal College of Midwives, Royal College of Obstetricians and Gynaecologists. RCM/RCOG consensus statement on Electronic Fetal Monitoring, Royal College of Midwives, London, 2017. (https://www.rcm.org.uk/media/2305/fetal-monitoring-consensus-statement.pdf) (Available from:).
- [25] Royal College of Midwives, Care Outside Guidance: Caring for Those Women Seeking Care That Falls Outside Guidance, RCM, London, 2022.
- [26] Royal College of Obstetrics Gynaecology, Birth After Previous Caesarean Birth. Green top guideline, RCOG, London, 2015, p. 45.
- [27] Royal United Hospitals Bath NHS Foundation Trust, Vaginal Birth After Caesarean Birth, Royal United Hospitals Bath NHS Foundation Trust, Bath, 2016. (https

- ://www.ruh.nhs.uk/patients/patient_information/MAT013_Vaginal_Birth_after_Ca esarean.pdf). Accessed online 20/8/23].
- [28] B. Saunders, J. Sim, T. Kingstone, et al., Saturation in qualitative research: exploring its conceptualization and operationalization, Qual. Quant. 52 (4) (2018) 1893–1907
- [29] V. Smith, C. Begley, D. Devane, Chapter 10 Technology in childbirth: Exploring women's views of fetal monitoring during labour a systematic review, in: S. Church, L. Frith, M.C. Balaam, M. Berg, V. Smith, C. van der Walt, S. Downe, E. van Teijlingen (Eds.), New thinking on improving maternity care: International perspectives, Pinter and Martin Ltd, London, 2017, pp. 170–193.
- [30] V. Smith, C. Begley, J. Newell, S. Higgins, D.J. Murphy, M.J. White, J.J. Morrison, S. Canny, D. O'Donovan, D. Devane, Admission cardiotocography versus intermittent auscultation of the fetal heart in low-risk pregnancy during evaluation for possible labour admission a multicentre randomised trial: the ADCAR trial, BJOG 126 (1) (2019) 114–121, https://doi.org/10.1111/1471-0528.15448.
- [31] R. Thompson, Y. Miller, Birth control: To what extent do women report being informed and involved in decisions about pregnancy and birth procedures? BMC Pregnancy Childbirth 14 (2014) 62. (http://www.biomedcentral.com/1471-2393/ 14/62)
- [32] University Hospitals Southampton (2019) (https://www.uhs.nhs.uk/Media/UHS-website-2019/Patientinformation/Pregnancyandbirth/Raised-BMI-in-pregnancy.pdf) [accessed online 10/11/23] University Hospitals Coventry and Warwickshire [accessed online 10/11/23] (https://www.birthability.co.uk/updated-vbac-guidelines/).
- [33] S. Villarmea, B. Kelly, Barriers to establishing shared decision-making in childbirth: unveiling epistemic stereotypes about women in labour, J. Eval. Clin. Pr. 26 (2020) 515–519, https://doi.org/10.1111/jep.13375.
- [34] J.V. Welsh, A.G. Symon, Unique and proforma birth plans: a qualitative exploration of midwives' experiences, Midwifery 30 (7) (2014) 885–891, https://doi.org/10.1016/j.midw.2014.03.004.
- [35] World Health Organization. Intrapartum care for a positive childbirth experience Geneva: World Health Organization; 2018. Available from: (https://www.who.int/reproductivehealth/publications/intrapartum-care-guidelines/en/).
- [36] C. Yuill, C. McCourt, H. Cheyne, et al., Women's experiences of decision-making and informed choice about pregnancy and birth care: a systematic review and meta-synthesis of qualitative research, BMC Pregnancy Childbirth 20 (2020) 343, https://doi.org/10.1186/s12884-020-03023-6.